Author’s response to reviews

Title: Use of GP services by patients with schizophrenia compared to patients with diabetes - a national cross-sectional register-based study

Authors:

Øystein Hetlevik (oystein.hetlevik@isf.uib.no)
Magne Solheim (magne.solheim@k2.uib.no)
Sturla Gjesdal (sturla.gjesdal@igs.uib.no)

Version: 2; Date: 28 March 2014

Author’s response to reviews:

Editorial office
BMC Health Service Research

Dear Editor.

We now submit a revised version of MS: 1122914653125302 - Use of GP services by patients with schizophrenia compared to patients with diabetes - a national register-based study.

Your comments are addressed below, with number referring to your e-mail.

1. Title page:
We have included the description of the study type “cross-sectional” in the title, and changed the title to: Use of GP services by patients with schizophrenia compared to patients with diabetes - a national cross-sectional register-based study.

The information about the authors is changed, hopefully now in line with your standard.

2. Acknowledgements
Concerning the section “acknowledgements” we have in this paper considered it most correct not to fill in anything here.

3. Text formatting:
Line numbering has been added. We have also changed the format and fonts on headings and subheadings in line with the BMC template.

Please contact us if there is something we have not made right.

Bergen 28.03.2014

Kind regards

Øystein Hetlevik